Author’s response to reviews

Title: Organizational culture, team climate and diabetes care in small office-based practices

Authors:

Marije Bosch (m.bosch@iq.umcn.nl)
Rob Dijkstra (r.dijkstra@nhg.org)
Michel Wensing (m.wensing@iq.umcn.nl)
Trudy van der Weijden (trudy.vanderweijden@hag.unimaas.nl)
Richard Grol (r.grol@iq.umcn.nl)

Version: 3 Date: 6 August 2008

Author’s response to reviews:

Dear Sir, Madam,

Thank you very much for accepting our paper for publication in BMC HSR. Please find the final version in which we have changed the following:

* as requested, we deleted the figure placement text, and cited the figure in the text
* as requested, we checked the references since there was a discrepancy between our references and the number of references recognized by your system. This should be solved now.
* we also replaced 's-spelling' with 'z-spelling', and made some final minor corrections.

Thank you very much on behalf of all authors,

Yours sincerely,

Marije Bosch